Author's response to reviews

Title: Development of a minimization instrument for allocation of a hospital-level performance improvement intervention to reduce waiting times in Ontario Emergency Departments.

Authors:

Chad A Leaver (chad.leaver@ices.on.ca)
Astrid Guttmann (astrid.guttmann@ices.on.ca)
Merrick Zwarenstein (merrick.zwarenstein@ices.on.ca)
Brian H Rowe (brian.rowe@ualberta.ca)
Geoff Anderson (geoff.anderson@ices.on.ca)
Therese Stukel (stukel@ices.on.ca)
Brian Golden (brian.golden@rotman.utoronto.ca)
Robert Bell (Robert.Bell@uhn.on.ca)
Dante Morra (dante.morra@utoronto.ca)
Howard Abrams (Howard.Abrams@uhn.on.ca)
Michael J Schull (mjs@ices.on.ca)

Version: 5 Date: 22 May 2009

Author's response to reviews:

Dear Editor,

We have addressed all the major and minor revisions outlined in your letter dated May 19th. We have also taken the opportunity to address some very minor text edits at this time.

On behalf of my coauthors and Dr. Schull, thank you very much for the opportunity for this work to appear in Implementation Science.